Author's response to reviews

Title: Household disaster preparedness and information sources: Rapid cluster survey after a storm in New South Wales, Australia

Authors:

Michelle A Cretikos (mcretikos@optusnet.com.au)
Keith Eastwood (keith.eastwood@hnehealth.nsw.gov.au)
Craig Dalton (craig.dalton@hnehealth.nsw.gov.au)
Tony Merritt (tony.merritt@hnehealth.nsw.gov.au)
Frank Tuyl (frank.tuyl@hnehealth.nsw.gov.au)
Linda Winn (linda.winn@hnehealth.nsw.gov.au)
David N Durrheim (david.durrheim@hnehealth.nsw.gov.au)

Version: 3 Date: 10 May 2008

Author's response to reviews:

Dear Claudia

We have made the requested changes to the manuscript including:
* providing details of authors contributions
* revision of the title page to conform with the journal's requirements
* revision of the format of the manuscript in general to conform with the journal's requirements.

I hope that these changes are satisfactory. The revised manuscript has been uploaded.

Kind regards,
Dr Michelle Cretikos